Dear Editors and Reviewers

Thank you for giving us the opportunity to submit a revised draft of the manuscript “Pulmonary lymphomatoid granulomatosis in a 4-year-old girl: A case report and literature review”

(Manuscript NO.: 71381) for publication in the Journal of World Journal of Clinical Cases.

Those comments are valuable and very helpful for revising and improving our paper. We appreciate the time and effort that you and the reviewers dedicated to providing feedback on our manuscript and are grateful for the insightful comments on and valuable improvements to our paper.

We have incorporated most of the suggestions made by the reviewers. Those changes are highlighted within the manuscript. Please see below, for a point-by-point response to the reviewers’ comments and concerns. All page numbers refer to changes in the revised manuscript file.

Response to comment:

Reviewer #1:

Comments: Specific Comments to Authors: Please include the FACS result of CD20[+], CD30[+], and CD3[+]

Answer: Thank you very much for the suggestions given by the reviewer. When the biopsy was made we had no idea what the lesions were, pulmonary lymphomatoid granulomatosis was not suspected, so Fluorescence Activating Cell Sorter (FACS) test was not performed, so we cannot provide relevant results and graphics. However, if we encounter similar cases in the future, we will consider adding relevant examinations to make the diagnosis more comprehensive.
Reviewer #2:

Some minor suggestions for revision:

**Comments:** 1. Detail the follow-up period - which chemotherapy regimen was she prescribed?

**Answer:** Thanks for the suggestion. The patient received a chemotherapy regimen of prednisone, vincristine and cyclophosphamide. After that she received bone marrow transplantation. We add this information in the manuscript (Line 187-189).

**Comments:** 2. Please check that all paragraphs have references - please check the Discussion section in particular.

**Answer:** Thanks for the suggestion. We have rechecked the manuscript and add references (Line 243-254). Now all paragraphs have references.

**Comments:** 3. The quality of the Figures is bad. Try to improve them, please.

**Answer:** Thanks for the suggestion. We improve the quality of the figures, and now each figure in the manuscript is 300dpi.

Reviewer #3:

**Comments:** 1. Please revise the English language and flow, and check throughout the text for spelling errors.

**Answer:** Thank you for pointing this out. We re-read the whole document, and there were indeed spelling errors. After the revision, we again asked a language retouching company to retouch the article to ensure that the language quality meets the requirements of the journal.

**Comments:** 2. I would suggest to add more recent literature (papers published in the last three years).

**Answer:** Thanks for the suggestion. Since the incidence of pulmonary lymphomatoid granulomatosis is very low in children, articles on this topic published in recent years are very few. We have added a selection of references on the chest CT features of the disease (reference 16 and 17, line 336-341).
Kind regards.

Mr Yao

E-mail：yaojw136138@163.com

Corresponding author：Lina Chen

E-mail address：chenlina_78@163.com